Author’s response to reviews

Title: Detection of varicella-zoster virus from cerebrospinal fluid using advanced fragment analysis in a child with encephalitis: A case report

Authors:
Yu-xin Song (635195698@qq.com)
Ye Li (416676827@qq.com)
Yong-mei Jiang (jiangym_SCU@163.com)
Ting Liu (liuting88823@163.com)

Version: 3 Date: 09 Apr 2019

Author’s response to reviews:

Response to the comments of the editor

To the specific comment 1: “We note the addition of a new author, [Yong-mei Jiang], since the original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts. We would therefore need to ask that all authors complete our change of authorship form”.

According to the requirements, we have completed the change of authorship form and returned it to BMCSeriesEditorial@biomedcentral.com.

To the specific comment 2: “Thank you for providing a response to the reviewers/CARE document. As these documents are no longer required at this stage of the publication process, please remove them from your submission’s supplementary files.”.

Thank you very much for your comments. According to the requirements, we have removed the point-by-point response letter and CARE document from submission’s supplementary files.

To the specific comment 3: “At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.”.

Thank you very much for your comments. According to the requirements, we have re-uploaded a clean version of manuscript.